Author's response to reviews

Title: Analysis of Network Use Behavior of Clinical Medical Students in China

Authors:

Hua yang (yangh@sj-hospital.org)
CHEN Yunxiang (chenyx@sj-hospital.org)
ZHENG Liqiang (zhenglq@sj-hospital.org)
XU Xin (xux@sj-hospital.org)
CAO Xia (caox@sj-hospital.org)
SHI Shuxia (shisx@sj-hospital.org)
ZHANG Jinghai (zhangjh2@sj-hospital.org)
ZHANG Fang (tusg2@sj-hospital.org)
YU Guang (yug@sj-hospital.org)
ZHAO Yueyang (zhaoyy@sj-hospital.org)

Version: 2
Date: 13 December 2013

Author's response to reviews: see over
Cover letter

By submitting the manuscript to the Journal, the authors understand that the material presented in this paper has not been published before nor has it been submitted for publication to another scientific journal or being considered for publication elsewhere. This manuscript is original and it, or any part of it, has not been previously published; nor is it under consideration for publication elsewhere. I attest that this work has been approved by all co-authors. The authors also understand that should the submitted material be accepted for publication in the journal, they will automatically transfer the copyright to the publisher. None of the authors has any conflict of interest, financial or otherwise. Each of the authors has contributed to, read and approved this manuscript.

I hope that the reviewing process finds the manuscript acceptable for publication in the journal.

Corresponding author:
YANG Hua
Library of Shengjing Hospital, China Medical University Shenyang 110004, Liaoning Province, China
Tel: +86-18940251151
Email: yangh@sj-hospital.org